Author's response to reviews

Title: Cost analysis of Multiple Sclerosis in Brazil: a cross-sectional multicenter study

Authors:

Nilceia Lopes da Silva (nilceia.lopes@novartis.com)
Maira L S Takemoto (maira.takemoto@anova.org.br)
Alfredo Damasceno (damascen@unicamp.br)
Yara D Fragoso (yara@bsnet.com.br)
Alessandro Finkelsztajn (alessandro.finkels@gmail.com)
Jefferson Becker (jeffersonbecker@hotmail.com)
Marcus V M Goncalves (marcusribeirao@yahoo.com.br)
Charles Tilbery (charlespt@einstein.br)
Enedina M L Oliveira (enedina.oliveira@unifesp.br)
Dagoberto Callegaro (dgcallegaro@yahoo.com)
Fernanda C Boulos (fernanda.boulos@novartis.com)

Version: 11 Date: 24 February 2016

Author's response to reviews: see over
Dear Editor,

Thank you for the peer review process. We have made the editorial request, as described below, and also some spelling corrections.

**Editorial request:**

- Please add the full names and affiliations of all the ethics committees that approved the study to the manuscript.

R.: Modified as proposed.